Author's response to reviews

Title: Deletions in the Y-derived amelogenin gene fragment in the Indian population

Authors:

V K Kashyap (vkk2k@hotmail.com)
Sanghamitra Sahoo (sanghamitra_sahoo2001@yahoo.com)
T Sitalaximi (ssita2k@yahoo.com)
R Trivedi (trivedi_r@hotmail.com)

Version: 3 Date: 30 March 2006

Author's response to reviews:

To,
Dr Sally Simpson,
Assistant Editor,
BMC Medical Genetics
BioMed Central

Subject: Submission of Revised Manuscript (# MS ID: 1544536015848305)

Dear Dr Simpson,

Thank you for accepting our manuscript entitled "Deletions in the Y-derived amelogenin gene fragment in the Indian population" for publication in BMC Medical Genetics. The minor essential revisions as suggested by the reviewers and formatting changes advised by the journal have been duly incorporated in the revised text.

We now hope that you would find the paper in the revised form suitable for publication in BMC Medical Genetics.

Looking forward to a positive response.

Thank You Once Again,
Sincerely,
V K Kashyap